Author's response to reviews

Title: A PILOT STUDY ON THE USEFULNESS OF PERIPHERAL BLOOD FLOW CYTOMETRY FOR THE DIAGNOSIS OF LOWER RISK MYELODYSPLASTIC SYNDROMES: THE "MDS THERMOMETER"

Authors:
Ana Aires (aninha_aires@hotmail.com)
Maria dos Anjos Teixeira (coutinhoteixeira@gmail.com)
Catarina Lau (mcatarinalau@gmail.com)
Marta Gonçalves (mrtcgon@gmail.com)
Cláudia Moreira (claudia.s.p.moreira@gmail.com)
Ana Spínola (a.fraco.spinola@gmail.com)
Ana Sofia Jorge (a_sofia_j@hotmail.com)
Vanessa Mesquita (mesquita.a.vanessa@gmail.com)
Renata Cabral (renatacabral20@hotmail.com)
Alexandra Mota (alexandra.mota@sapo.pt)
Cristina Gonçalves (cristina.p.goncalves1@gmail.com)
Luciana Pinho (lpinho48@hotmail.com)
Luciana Xavier (luciana@imania.pt)
Inês Freitas (maria.ines.freitas@hotmail.com)
Jorge Coutinho (jqcoutinho@yahoo.com)
Margarida Lima (mmc.lima@clix.pt;margaridalima@chporto.min-saude.pt)

Version: 3 Date: 10 Dec 2017

Author's response to reviews:

BHEM-D-17-00045: Revised version 3

A PILOT STUDY ON THE USEFULNESS OF PERIPHERAL BLOOD FLOW CYTOMETRY FOR THE DIAGNOSIS OF LOWER RISK MYELODYSPLASTIC SYNDROMES: THE "MDS THERMOMETER"
10-12-2017

To the Editorial Office of BMC Hematology

Dear Editor,

Dear Reviewers,

According to your indications, I am sending a clean version of the manuscript BHEM-D-17-00045 (R3), where, as suggested, I made the following changes, compared to the last version (R2):

a) The list of authors was reviewed, taking in account the guidelines regarding qualification for authorship. In accordance, MG, ASJ, CG, LP, LX, RC, and VM were removed from the list of authors and placed in the “Acknowledgements” section (please see Title page - page 1, Authors' contributions – page 24; Acknowledgements – page 25, and Authors' information – page 25).

b) The full name of the ethics committee that approved the study and the institute to which it belongs to was added; the study reference number was already indicated (please see Ethics approval and consent to participate – page 24).

We are completing the “Change in authorship” form and I will send it as soon as possible (two of the contributors are not in Portugal for the moment).

Hoping that the manuscript deserves the final approval for publication in BMC Hematology,
Yours sincerely

Margarida Lima

Corresponding author: Margarida Lima, Laboratório de Citometria, Serviço de Hematologia, Hospital de Santo António, Centro Hospitalar do Porto, instalações do Ex-CICAP, Rua D. Manuel II, s/n, 4099-001 Porto, Portugal; @-mail: mmc.lima@cliix.pt; Telephone number: + 351 22 20 77 500